Author’s response to reviews

Title: Development and Validation of a Prediction Model for Knee Joint Line Orientation after High Tibial Osteotomy

Authors:
Jae-Young Park (neoxcv@gmail.com)
Chong Bum Chang (ccbknee@gmail.com)
Dong-Wan Kang (kndnwn@gmail.com)
Sohee Oh (oh.sohee@gmail.com)
Seung-Baik Kang (ssbkang@snu.ac.kr)
Myung Lee (leemc@snu.ac.kr)

Version: 2 Date: 04 Sep 2019

Author’s response to reviews:

Assistant Editor Comments:

1. Overlap
We note that the current submission contains some textual overlap with other previously published works, in particular:


While we understand that this is work that you have previously published, and some of the same ideas are contained in these publications, please be aware that we cannot condone the use of text from previously published work.
Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated.

Answer:

Thank you for the comment. We have resolved the issue of textual overlap. Especially the overlap with the previous publication by Lee, Kyoung Min has been eliminated. Thank you.

2. Abbreviations

Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Answer:

We have added abbreviations as you have commented, thank you.

3. Acknowledgements

We note that you have not included an acknowledgements section in your Declarations section. If you have no acknowledgements please put ‘Not Applicable’ in this section.

Answer:

We have added acknowledgements as you have commented, thank you.

4. Consent for publication

We note that you have not included a ‘Consent for publication’ section in Declarations. If identifying images or other personal or clinical details of participants are presented that compromise anonymity, a statement of consent to publish from the patient should be included. This section must be included even if it is not applicable to your manuscript. If consent to publish is not applicable to your manuscript please write ‘Not Applicable’ in this section.
We note that you have possibly identifying information (age and gender) in figure legend 3. If you do not have consent to publish please replace this with “a patient”.

Answer:

We have added consent for publication as you have commented, thank you. We have also changed the figure legend 3 to “a patient” as you have commented, thank you.

5. Supplementary material

Please remove the IRB approval evidence from the file inventory as it is no longer needed at this stage.

Answer:

We have removed it.

6. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Answer:

We have uploaded a clean version, thank you.